Author's response to reviews

Title: Chronic pain self-management for older adults: a randomized controlled trial

Authors:

Mary Ersek (mary.ersek@swedish.org)
Judith A Turner (jturner@u.washington.edu)
Kevin C Cain (cain@u.washington.edu)
Carol A Kemp (carola.kemp@swedish.org)

Version: 2 Date: 26 July 2004

Author's response to reviews:

We have made the formatting changes requested in the revised manuscript.
Thank you.
Judith A. Turner, Ph.D.